Title: Chronic hepatitis E virus infection in a leukemia patient with elevated transaminases: a case report

Authors:

Annika Braun (Annika.Braun@med.uni-heidelberg.de)
Jürgen J. Wenzel (Juergen.Wenzel@klinik.uni-regensburg.de)
Christa Flechtenmacher (Christa.Flechtenmacher@med.uni-heidelberg.de)
Mojdeh Heidary Navid (Mojdeh.Heidary-Navid@med.uni-heidelberg.de)
Christoph Eisenbach (Christoph.Eisenbach@med.uni-heidelberg.de)
Wolfgang Jilg (Wolfgang.Jilg@klinik.uni-regensburg.de)
Wolfgang Stremmel (Wolfgang.Stremmel@med.uni-heidelberg.de)
Paul Schnitzler (Paul_Schnitzler@med.uni-heidelberg.de)

Version: 4 Date: 14 August 2012

Author’s response to reviews: see over
Dear Prof. Kidd,

thank you very much for kindly reviewing our manuscript entitled

"Chronic hepatitis E virus infection in a leukemia patient with elevated transaminases: a case report"
by A. Gauss, J.J. Wenzel, C. Flechtenmacher, M. Heidary Navid, C. Eisenbach, W. Jilg, W. Stremmel, P. Schnitzler

Enclosed please find our revised manuscript for your kind reconsideration for publication in "Journal of Medical Case Reports". The English grammar and language have been revised throughout the manuscript, all corrections are marked in red.

We hope that our manuscript is now in a suitable form for publication in "Journal of Medical Case Reports".

With kindest regards

Dr. Paul Schnitzler